Author's response to reviews

Title: The Improved Delivery of Cardiovascular Care (IDOCC) through Outreach Facilitation: study protocol and implementation details

Authors:

Clare Liddy (cliddy@bruyere.org)
William Hogg (whogg@uottawa.ca)
Grant Russell (grant.russell@med.monash.edu.au)
George Wells (gawells@ottawaheart.ca)
Catherine Deri Armstrong (cdarmstrong@uottawa.ca)
Ayub Akbari (aakbari@ottawahospital.on.ca)
Simone Dahrouge (sdahrouge@bruyere.org)
Monica Taljaard (mtaljaard@ohri.ca)
Liesha Mayo-Bruinsma (lmayobruinsma@bruyere.org)
Jatinderpreet Singh (jsingh@bruyere.org)
Alex Cornett (acornett@bruyere.org)

Version: 2 Date: 30 March 2011

Author's response to reviews: see over
Dear Mr. Wilson,

Thank you for your response to the submission of our manuscript which is now entitled: “The Improved Delivery of Cardiovascular Care (IDOCC) through Outreach Facilitation: study protocol and implementation details of a cluster randomized control trial in primary care.” We have revised the study protocol to address the issues you raised.

A CONSORT 2010 checklist is included in the submission package. We changed the title of the protocol to include “randomized controlled trial”. We added a list of abbreviations to the main text. The figures are sent in separate files. We are in the process of obtaining the permission to use a figure from American College of Physicians which is included in the protocol as Figure 1. A table (Table 3) was added which shows the progress of the project with dates and the fact that the trial is not yet completed.

Thank you again for considering our article for publication.

Sincerely,

Clare Liddy, MD, MSc. CCFP, FCFP
Assistant Professor,
Department of Family Medicine
University of Ottawa
Clinical Investigator,
C.T. Lamont Primary Health Care Research Centre
Élisabeth Bruyère Research Institute
43 Bruyère St. Room 369Y
Ottawa (Ontario) K1N 5C8
Tel: 613-562-6262, x1514
Fax: (613) 562-6099
CLiddy@bruyere.org